University of Texas Rio Grande Valley

ScholarWorks @ UTRGV

Earth, Environmental, and Marine Sciences Faculty Publications and Presentations

College of Sciences

6-2020

Potential Predictability of Net Primary Production in the Ocean

Kristen M. Krumhardt

Nicole S. Lovenduski

Matthew C. Long

J. Y. Luo

K. Lindsay

See next page for additional authors

Follow this and additional works at: https://scholarworks.utrgv.edu/eems_fac

Part of the Earth Sciences Commons, Environmental Sciences Commons, and the Marine Biology Commons

Recommended Citation

Krumhardt, K. M., Lovenduski, N. S., Long, M. C., Luo, J. Y., Lindsay, K., Yeager, S., & Harrison, C. (2020). Potential predictability of net primary production in the ocean. Global Biogeochemical Cycles, 34, e2020GB006531. https://doi.org/10.1029/2020GB006531

This Article is brought to you for free and open access by the College of Sciences at ScholarWorks @ UTRGV. It has been accepted for inclusion in Earth, Environmental, and Marine Sciences Faculty Publications and Presentations by an authorized administrator of ScholarWorks @ UTRGV. For more information, please contact justin.white@utrgv.edu, william.flores01@utrgv.edu.

Authors

Kristen M. Krumhardt, Nicole S. Lovenduski, Matthew C. Long, J. Y. Luo, K. Lindsay, S. Yeager, and Cheryl S. Harrison



Global Biogeochemical Cycles

RESEARCH ARTICLE

10.1029/2020GB006531

Key Points:

- Marine net primary production is potentially predictable for at least 1 year in advance for many oceanic regions
- Regions where nutrients are the primary driver of net primary production are more predictable than light- or temperature-driven regions
- The dynamic prediction system improves predictability of net primary production in some Large Marine Ecosystems over a persistence forecast

Supporting Information:

• Supporting Information S1

Correspondence to: K. M. Krumhardt,

kristenk@ucar.edu

Citation:

Krumhardt, K. M., Lovenduski, N. S., Long, M. C., Luo, J. Y., Lindsay, K., Yeager, S., & Harrison, C. (2020).Potential predictability of net primary production in the ocean. *Global Biogeochemical Cycles*, *34*, e2020GB006531. https://doi.org/10. 1029/2020GB006531

Received 5 FEB 2020 Accepted 6 MAY 2020 Accepted article online 8 JUN 2020

Potential Predictability of Net Primary Production in the Ocean

K. M. Krumhardt^{1,2}, N. S. Lovenduski³, M. C. Long², J. Y. Luo^{2,4}, K. Lindsay², S. Yeager², and C. Harrison⁵

¹Environmental Studies Program and Institute of Arctic and Alpine Research, University of Colorado Boulder, Boulder, CO, USA, ²Climate and Global Dynamics, National Center for Atmospheric Research, Boulder, CO, USA, ³Department of Atmospheric and Oceanic Sciences and Institute of Arctic and Alpine Research, University of Colorado Boulder, Boulder, CO, USA, ⁴Geophysical Fluid Dynamics Laboratory, NOAA, Princeton, NJ, USA, ⁵Port Isabel Laboratory, University of Texas Rio Grande Valley, Edinburg, TX, USA

Abstract Interannual variations in marine net primary production (NPP) contribute to the variability of available living marine resources, as well as influence critical carbon cycle processes. Here we provide a global overview of near-term (1 to 10 years) potential predictability of marine NPP using a novel set of initialized retrospective decadal forecasts from an Earth System Model. Interannual variations in marine NPP are potentially predictable in many areas of the ocean 1 to 3 years in advance, from temperate waters to the tropics, showing a substantial improvement over a simple persistence forecast. However, some regions, such as the subpolar Southern Ocean, show low potential predictability. We analyze how bottom-up drivers of marine NPP (nutrients, light, and temperature) contribute to its predictability. Regions where NPP is primarily driven by the physical supply of nutrients (e.g., subtropics) retain higher potential predictability than high-latitude regions where NPP is controlled by light and/or temperature (e.g., the Southern Ocean). We further examine NPP predictability in the world's Large Marine Ecosystems. With a few exceptions, we show that initialized forecasts improve potential predictability of NPP in Large Marine Ecosystems over a persistence forecast and may aid to manage living marine resources.

Plain Language Summary Marine net primary production (NPP) is the base of the marine food web, as well as an important component of the ocean carbon cycle. Year-to-year variations in NPP can influence the availability of living marine resources, such as fish. In this study, we show that an Earth System Model can be used to generate near-term (1 to 10 years) forecasts of marine NPP. Earth System Model-based forecasts of NPP show an improvement over a persistence forecast, where NPP the following year is assumed to be the same as the current year's NPP. Annual NPP variations can be predicted for 1 to 3 years in advance in many oceanic regions, from temperate waters to the tropics. NPP in colder regions, however, is harder to predict. The main drivers of NPP influence its predictability. Places where nutrient availability primarily drives variations in NPP are more predictable than regions of the ocean where light and temperature are the main drivers (primarily high latitude regions). We further demonstrate NPP predictability in coastal regions, the world's Large Marine Ecosystems. We show that NPP predictions could be potentially useful in many Large Marine Ecosystems, and this may help to sustainably manage coastal marine ecosystems.

1. Introduction

Net primary production (NPP) by ocean phytoplankton provides energy to marine ecosystems. Fluctuations in oceanic NPP can therefore lead to variations in living marine resources (Pauly & Christensen, 1995). Not only is marine NPP ecologically important, but it is also a major component of the global carbon cycle, as marine phytoplankton are responsible for roughly half of carbon fixed through photosynthesis each year on a global scale (Falkowski, 2012). A portion of the carbon fixed through NPP in the surface ocean sinks to depth, concentrating carbon in the deep ocean. NPP can vary significantly from year to year, as bottom-up environmental factors controlling phytoplankton NPP (e.g., nutrients, light, and temperature) are subject to long-term trends, as well as substantial interannual variability (Krumhardt et al., 2017; Séférian et al., 2014). While NPP is the ultimate control on net ecosystem production, upper trophic level biomass is not always related to NPP in a straightforward way (see, e.g., Friedland et al., 2012;

©2020. American Geophysical Union. All Rights Reserved. Sherman et al., 2009; Stock et al., 2017). Nevertheless, making near-term (1 to 10 years) forecasts of marine NPP could contribute to more effective management of living marine resources (e.g., Lotze et al., 2019; Payne et al., 2017; Tommasi et al., 2017), improve conservation efforts in marine protected areas (Edgar et al., 2014), and inform climate change adaptation strategies.

Recent progress has been made in generating near-term ("decadal") predictions of physical and biogeochemical quantities using Earth system models (ESMs; see, e.g., Meehl et al., 2014). Initialized ESM forecasts have proven useful for predictions of ocean heat content (Yeager et al., 2012, 2018), sea surface temperature (SST) and air surface temperature (Smith et al., 2007; Stock et al., 2015; Yeager et al., 2018), ocean carbon uptake (Li et al., 2019; Lovenduski et al., 2019), precipitation (Yeager et al., 2018), and sea ice (Yeager et al., 2015) on timescales ranging from 1 to 10 years in advance. Most recently, Park et al. (2019) described seasonal to 2-year predictability of ocean chlorophyll concentration and demonstrate connections with annual fish catch data, illustrating the feasibility and utility of ESM predictions for marine ecosystem quantities. While chlorophyll concentration is subject to substantial variability from phytoplankton physiological changes that may not be indicative of phytoplankton biomass (Behrenfeld et al., 2015), NPP quantifies the net rate of carbon fixation by phytoplankton, an important quantity for marine ecosystem modeling (Lotze et al., 2019). Yet the predictability of NPP on decadal timescales has been explored in relatively few studies. Séférian et al. (2014) demonstrated 3-year predictability of NPP in the tropical Pacific Ocean, surpassing predictability of SST and showing that nutrient advection plays an important role for multiyear prediction of NPP. Further, Yeager et al. (2018) also showed that ESM-based decadal prediction system holds promise in predicting marine NPP in numerous ocean regions, including the tropical and subtropical Atlantic and Pacific, and some eastern boundary upwelling systems, leading to questions about what enables predictability in NPP.

An important requirement for forecasting interannual variations in marine NPP is adequate prediction of its drivers. Bottom-up drivers of NPP include temperature, nutrient availability, and light, while grazing constrains NPP in a top-down sense (Laufkötter et al., 2015). Each of these NPP drivers is influenced by internal climate variability (Krumhardt et al., 2017); the extent to which NPP is predictable may depend on the predictability of its primary drivers. SST, which influences phytoplankton metabolic rates, has been shown to be potentially predictable on wide spatial scales for up to 9 years, with the exception of the Southern Ocean (Yeager et al., 2018). Though the persistence of nutrient anomalies in the tropical Pacific was shown by Séférian et al. (2014) to be an important driver of NPP, the predictability of nutrient advection, a potentially large driver of NPP in much of the ocean, has not been demonstrated using a dynamic ESM-based prediction system.

While predictions of NPP in open ocean regions are ecologically and biogeochemically important, fluctuations of NPP in coastal regions may be of higher interest for society. The world's Large Marine Ecosystems (LMEs) comprise oceanic areas on which humans are most dependent; roughly 90% of living marine resources are harvested within the LMEs (Pauly et al., 2008; Sherman, 2005). LMEs are characterized by distinct bathymetry, hydrography, productivity, and trophically dependent populations. As such, these regions are seen as effective units for marine ecosystem management (Sherman, 2014). Additionally, LMEs contain a majority of marine protected areas which are seen as primary management regions for mitigating the threats of climate change and other human influences on marine biodiversity (Bruno et al., 2018). Predictions of NPP in these regions may aid to effectively manage LMEs with the goals of sustainability and conservation (Edgar et al., 2014).

Forecasts of physical and ecological variables using dynamical prediction systems are commonly compared to a null "persistence" forecast, which assumes that the following year will be same as the current year (lagged autocorrelation; e.g., see Stock et al., 2015; Yeager et al., 2018). A persistence forecast therefore provides a baseline with which to describe the benefit of using a more complex prediction system. While a persistence forecast may be adequate for certain variables or regions that show longer timescales of variability, those that have strong interannual fluctuations (like marine NPP in most regions) would potentially benefit most from a dynamical prediction system.

In addition to comparing to persistence, there are also two different types of forecast predictability measures. "Skillful prediction" refers to the ability of a forecast to match observations. For instance, Park et al. (2019) assess skillful prediction of chlorophyll by comparing to satellite-derived estimates of ocean chlorophyll content. In contrast, "potential predictability" assesses forecasts of ocean variables using a historical reconstruction of the ocean state, generated by an ESM simulation forced by atmospheric observations. This allows full field initialization of the ESM for each forecast and evaluation of forecasts over the entirety of the historical reconstruction, rather than being limited to the satellite era. Given the uncertainty and variation across satellite-derived NPP estimates (e.g., see Saba et al., 2011), we focus our analysis on potential predictability of NPP in the global oceans in this study, comparing NPP forecasts to a ESM-based historical reconstruction of marine NPP.

We use a large ensemble of initialized forecasts from the Community Earth System Model (CESM) to demonstrate that interannual variations in NPP are potentially predictable for 1 to 3 years in many regions of the ocean, showing an improvement over a simple persistence forecast. By using a large ensemble of forecasts (each ensemble member differs by round-off level differences in initialization; see section 2), we are able to derive the most statistically probable forecast, the ensemble mean. We examine the main drivers of marine NPP in the global ocean in order to understand why NPP is more predictable in some regions than in others. We further focus on NPP predictability within the world's LMEs in order to assess potential predictability in societally relevant regions. We show that NPP forecasts in some coastal LMEs still offer a substantial improvement over persistence. This study highlights the potential for dynamic ESM-based prediction systems to forecast critical ecosystem metrics like marine NPP.

2. Materials and Methods

2.1. CESM Decadal Prediction System

We use the CESM Decadal Prediction Large Ensemble (CESM-DPLE Yeager et al., 2018) to explore potential predictability of marine NPP. This prediction system is described in detail elsewhere (Lovenduski et al., 2019; Yeager et al., 2018); we thus provide only a brief overview here. The CESM-DPLE consists of 40 forecasts starting each year from 1955 to 2016. Forecasts are initialized in November of the preceding year and then run for 10 full calendar years; therefore, a lead Year 1 forecast refers to the January through December year following initialization. The ensemble of forecasts produced for each year only differ by round-off level $(10^{-14\circ}C)$ differences in the initial air temperature field, which is sufficient to rapidly generate spread across the ensemble.

The code base for the CESM-DPLE is CESM version 1.1 with ocean biogeochemistry (Long et al., 2013; Moore et al., 2004, 2013) run fully coupled at a nominal 1° resolution. Initial conditions for the land and atmosphere were initialized using a single member of the CESM large ensemble (Kay et al., 2015) and, thus, were not constrained by observations. The ocean and sea ice, however, were initialized using a CESM forced ocean-sea ice (CESM-FOSI) simulation (including marine biogeochemistry) that was forced by historical atmospheric state and flux fields based on the Coordinated Ocean-Ice Reference Experiment reanalysis data set (Yeager et al., 2018); this setup allows full field initialization of physical and biogeochemistry ocean variables. In the section below, we evaluate this CESM-FOSI NPP reconstruction with respect to observationally based, satellite-derived NPP estimates. The forecast ensembles drift from their initialized state; thus, the variables from the CESM-DPLE presented here are all drift corrected by removing a forecast lead year-dependent model climatology from each variable, resulting in forecasted anomalies (see Yeager et al., 2018).

2.2. Data Processing and Statistics

We focus on the predictability of interannual anomalies in marine NPP integrated over the upper 150 m. We removed a linear trend from the forecasts and CESM-FOSI reconstruction in order to focus on the interannual variability apart from long-term trends over the study period. In this study, as in previous decadal prediction studies (e.g., Lovenduski et al., 2019; Yeager et al., 2018), we use anomaly correlation coefficients (ACC) between our observation-based initialization data set (here, the CESM-FOSI reconstruction) and the corresponding "forecasted" year from the CESM-DPLE to evaluate the "potential predictability" of variables in the Earth system. This comparison does not depend on model skill but rather evaluates the potential for initialized forecasts to predict the evolution of the system. We also note that ACC does not provide information on the magnitude of NPP anomalies but provides a measure of how variations in NPP match between forecasts and the CESM-FOSI reconstruction. Here, we use ACC as a measure of the relative association of the average forecast with the CESM-FOSI simulation. We compare NPP forecasts from the CESM-DPLE to a simple persistence forecast, which assumes that the following forecast period will be same as the current forecast period (i.e., autocorrelation). We use a *z*-test to evaluate the statistical difference in ACC at the 95% level between the persistence forecast and the CESM-DPLE forecast. We assess the influence of bottom-up drivers on NPP using variables from the CESM-FOSI reconstruction. These include the limitation terms for temperature, light, and nutrients, which modify the maximum growth rates of the phytoplankton functional types (PFTs) in the model. Each of these limitation terms ranges from 0 to 1, with 0 being the most limiting and 1 indicating plentiful supply. We determined the most limiting nutrient for each PFT for each grid cell in the top 150 m; therefore, the nutrient limitation term in this analysis corresponds to the nutrient most limiting phytoplankton growth. We created biomass-weighted limitation terms, weighting each limitation term by the biomass of the PFT experiencing the limitation over the top 150 m of the water column. The limitation terms for temperature, light, and nutrients were then detrended and correlated with depth-integrated NPP anomalies. We interpret the correlation coefficient as a measure of the influence of each limitation term on NPP, noting that this does not directly take the magnitude of the influence into account.

We also analyzed the CESM-DPLE potential predictability of various bottom-up drivers of NPP, including SST, photosynthetically active radiation (PAR) at the surface, mixed layer depth, and the total physically mediated tendency (i.e., rate of change) of PO_4 in the upper 100 m, not including biological uptake/remineralization. Though PO_4 is not always the nutrient that limits phytoplankton growth in the model, we chose PO_4 rather than NO_3 to represent macronutrient flux because (1) unlike NO_3 it is not complicated by NOx fluxes from the surface, (2) its influence on phytoplankton NPP is more straightforward than NO_3 (nitrogen fixation/ammonia supply influence N availability), and (3) NO_3 and PO_4 are highly correlated (Figure S1 in the supporting information) so their physical fluxes are likely very similar in most regions.

2.3. Evaluating the CESM-FOSI NPP Reconstruction

To assess the skill of CESM to simulate observed NPP, we compare the CESM-FOSI NPP reconstruction to NPP derived from the Moderate Resolution Imaging Spectroradiometer (MODIS) over the time period 2003 (first full year of MODIS) to 2015 (last year of the FOSI reconstruction). There exist numerous algorithms for calculating NPP from ocean color data, which offer substantially different solutions for estimating ocean NPP (Saba et al., 2011). Rather than selecting one of these algorithms, we opted to use the mean of three prominent models: the Vertically Generalized Production Model (VGPM; Behrenfeld & Falkowski, 1997), the Eppley-VGPM (Eppley; Behrenfeld & Falkowski, 1997; Eppley, 1972; Morel, 1991), and the Carbon-based Productivity Model (CbPM; Behrenfeld et al., 2005; Westberry et al., 2008). To evaluate the CESM-FOSI reconstruction in simulating marine NPP, we compare the mean annual NPP. We also interpolated CESM NPP and satellite-derived NPP to a common 2° grid in order perform correlations of NPP variability and compare standard deviation over the satellite period between the satellite-derived NPP and depth-integrated NPP from the CESM-FOSI reconstruction.

3. Results and Discussion

In this section we first provide an evaluation of the the CESM-FOSI NPP reconstruction with respect to satellite-derived NPP. We then present a global overview of NPP predictability and evaluate how NPP drivers may influence its predictability. Lastly, we examine NPP predictability in the world's LMEs.

3.1. Evaluation of the CESM-FOSI Reconstruction of NPP

Here, we evaluate the CESM-FOSI reconstruction of NPP as a surrogate for NPP observations. We focus on the evaluation of NPP because other ocean variables in the CESM-FOSI reconstruction have been previously evaluated in the literature (e.g., Lovenduski et al., 2019; Yeager et al., 2018). Marine NPP in the CESM-FOSI reconstruction shows good correspondence to satellite-derived marine NPP, as both show similar geographic patterns in marine NPP (Figures 1a and 1b). Many areas of the ocean show a significant positive correlation between modeled NPP and the satellite-derived NPP over the satellite period (Figure 1c; red areas). The CESM-FOSI reconstruction, however, tends to underestimate the overall magnitude of the coastal NPP according to the satellite-derived observations, especially in Eastern Boundary Upwelling systems and in the western Indian Ocean (Figures 1a and 1b). The relatively coarse resolution of the model likely contributes to this bias. The model underestimates NPP variability in these same regions (pink regions on Figure 1d), while overestimating NPP variability in the subpolar Southern Ocean and central equatorial Pacific (green regions on Figure 1d). These are also the regions where we observe the highest mean absolute error in the CESM-FOSI reconstruction referenced to satellite-derived NPP (Figure S2).





(c) CESM–FOSI NPP reconstruction correlation with satellite–derived NPP

(d) CESM-FOSI NPP recontruction σ / NPP MODIS σ



Figure 1. A comparison of the CESM-FOSI NPP reconstruction and satellite-derived NPP (mean of VGPM, Eppley, and CbPM; see section 2) over the period 2003 to 2015. Panels (a) and (b) show annual average satellite-derived NPP and NPP from the FOSI reconstruction, respectively. Panel (c) shows the correlation coefficients between annual mean FOSI NPP reconstruction and satellite-derived annual mean NPP (mean of VGPM, Eppley, and CbPM; see section 2); only correlations at 90% significance are shown. The ratio in standard deviation (σ) of annual means between the FOSI NPP reconstruction and satellite-derived NPP is plotted in panel (d). Large Marine Ecosystem (LME) boundaries are shown by black lines.

3.2. Potential Predictability of Marine NPP: A Global Overview

For each year of the CESM-DPLE, 40 retrospective forecasts of annual marine NPP anomalies each provide a plausible trajectory for marine NPP, with the ensemble mean being the most probable forecast. In Figure 2a, drift-corrected forecasts are shown for four example years to illustrate this concept (ensemble members shown in light green with the ensemble mean in dark green). Globally integrated NPP anomalies are predictable 1 year in advance, as a lead Year 1 forecast time series corresponds well to the CESM-FOSI NPP reconstruction (Figure 2b). Indeed, the correlation between the two lines in Figure 2b is ~0.8, but this rapidly decreases for subsequent lead years (Figure 2c, dark green line). Though the CESM-DPLE global NPP anomaly forecast shows higher potential predictability than a persistence forecast for lead Years 1 and 2, it is not significantly higher than a persistence forecast at the 95% level (Figure 2c, yellow line). A geographical view of potential predictability offers more insight into regions where NPP anomalies may show a significant improvement over a persistence forecast.

In Figures 3a–3c, we present maps of the potential predictability (measured by ACC) of marine NPP for forecast Years 1, 3, and 5 (see Figure S3 for maps of all forecast lead years). The improvement in potential predictability over a simple persistence forecast is presented Figures 3d–3f, with stippling highlighting areas where the CESM-DPLE NPP prediction shows a significant improvement over persistence. Potential predictability in marine NPP is high for most areas of the ocean 1 year in advance (Figure 3a). NPP is especially predictable from temperate waters to the tropics. The subpolar and polar regions of the Southern Ocean show low potential predictability in NPP, as do some equatorial regions (e.g., northern Indian Ocean, far west equatorial Pacific). Areas where there is a significant improvement over persistence for a lead Year 1 NPP forecast are mainly in the Southern Hemisphere, as well as some regions in the subtropical and temperate regions of the North Atlantic and North Pacific (Figure 3d). However, there are areas where a persistence



Figure 2. Predictability of globally integrated marine NPP anomalies in the CESM-DPLE: Panel (a) shows initialization of 40 ensemble members (in light green) from the CESM-FOSI NPP reconstruction (in gray) for initialization years 1955, 1969, 1986, and 2001 (ensemble means are shown in dark green); panel (b) shows correspondence between NPP anomalies over time for the FOSI NPP reconstruction and the lead Year 1 forecast; panel (c) shows anomaly correlation coefficients (ACC) as a function of lead year for a persistence forecast and the CESM-DPLE forecast for global NPP anomalies. Red circles indicate that the ACC is significant at 95%.

forecast performs better than the CESM-DPLE (brown areas in Figures 3d-3f), such as the upwelling regions along the eastern boundary of the South Atlantic. As forecast lead time increases, potential predictability of marine NPP tapers off quickly for many areas of the ocean (Figures 3b and 3c). Some regions, however, show NPP potential predictability for lead times >1 year; those include the South Pacific, western Atlantic, and southern Indian Ocean.

In order for predictions of marine NPP to be applied in a management sense, NPP must be accurately simulated in the ESM and predictable. Yeager et al. (2018) showed a global view of "skillfull predictability" for a lead Year 1 NPP forecast using the CESM-DPLE (see their Figure 10d). Here, we evaluated the ability of CESM to simulate NPP when forced by an observational atmospheric forcing product (CESM-FOSI reconstruction; see section 3.1), examining "potential predictability" separately. Some oceanic regions do indeed show higher correspondence when compared to satellite-derived NPP than others. For example, CESM-FOSI-reconstructed marine NPP in subtropical regions correlates well with satellite-derived NPP (red



Figure 3. Potential predictability of marine NPP in the CESM-DPLE: Anomaly correlation coefficients (ACC) between the CESM-FOSI NPP reconstruction and NPP predicted by the CESM-DPLE for lead Years 1 (a), 3 (b), and 5 (c). Only significant correlations (95%) are shown; white areas denote nonsignificant correlations. Panels (d), (e), and (f) show the CESM-DPLE potential predictability improvement over a persistence forecast (Δ ACC), with stippled areas showing where there is a significant improvement over persistence at the 95% level . Large Marine Ecosystem (LME) boundaries are shown by black lines.

areas in Figure 1c) and shows similar levels of variability (Figure 1d). These subtropical regions also have high potential predictability 1 year in advance (Figure 3a), demonstrating the potential applicability of subtropical NPP predictions. Likewise, parts of the North Atlantic, Western Pacific, and Eastern Indian Ocean show adequate model skill (as evaluated using the the CESM-FOSI simulation; section 3.1) and potential predictability. However, some high-latitude regions, such as the subpolar and polar Southern Ocean, show adequate correspondence to satellite-derived NPP but low potential predictability (Figures 1 and 3), suggesting low ability to predict the evolution of NPP drivers. In the following section we explore why some regions may have better potential predictability using the CESM-DPLE forecasting system than others.

3.3. Drivers of NPP Affect Its Predictability

We expect that interannual variations in NPP will be potentially predictable if the bottom-up drivers exhibit high potential predictability. To understand which bottom-up drivers (temperature, nutrients, or light) have the greatest influence on variations in marine NPP, we correlate biomass-weighted phytoplankton limitation terms with depth-integrated marine NPP anomalies from the CESM-FOSI reconstruction (see section 2.2 for more details). From this analysis, we interpret that marine NPP is most controlled by the limitation terms that show the highest correlation with NPP in the CESM-FOSI reconstruction. We further classify regions as being primarily controlled by a single bottom-up driver based on highest correlation to NPP anomalies. Nutrient limitation appears to drive NPP variations in the low- to middle-latitude ocean regions (Figure 4a). Light and temperature limitations show similar geographic patterns for driving marine NPP (see Figures 4b)

(a) Nutrient limitation and NPP correlation



(b) Temperature limitation and NPP correlation



(c) Light limitation and NPP correlation



-1.0 -0.8 -0.6 -0.4 -0.2 0.0 0.2 0.4 0.6 0.8 1.0 Correlation coefficient

(d) Highest correlated phytoplankton limitation term



Figure 4. Correlations between annual mean NPP anomalies and biomass weighted limitation terms from the CESM-FOSI reconstruction: (a) nutrient limitation (most limiting nutrient; see section 2), (b) temperature limitation, and (c) light limitation (nonsignificant correlations are masked with white). All variables were annually averaged and detrended prior to doing correlations. Panel (d) shows the limitation term with the highest correlation coefficient with NPP ("main driver"), with white areas denoting places were none of the correlations were significant at 95%. Large Marine Ecosystem (LME) boundaries are shown by black lines.

Global Biogeochemical Cycles



Figure 5. Potential predictability (anomaly correlation coefficients, ACC, between CESM-DPLE ensemble mean and the CESM-FOSI reconstruction) for lead Years 1, 3, and 5 for four NPP drivers: PO₄ physical supply (panels (a), (b), and (c)), sea surface temperature (SST; panels (d), (e), and(f)), surface photosynthetically active radiation (PAR; panels (g), (h) and (i)), and mixed layer depth (panels (j), (k), and (l)). See supporting information for all lead years. Only significant correlations (95%) are shown; white areas denote nonsignificant correlations. Large Marine Ecosystem (LME) boundaries are shown by black lines.

and 4c): Variability in temperature and light explains most NPP variance in the high-latitude oceanic regions. Interannual variability in NPP in the North Atlantic and most of the Southern Ocean appears to be primarily driven by variations in light, while NPP in the Arctic, and some small regions in the subpolar Southern Ocean south of Africa and parts of the North Pacific are primarily influenced by variations in temperature (Figure 4d). Though increases in temperature, light, and nutrients all have a positive effect on NPP in CESM, negative correlations also result from this analysis (blue areas on Figures 4a–4c). This is due to anticorrelations between the drivers. For instance, deeper mixed layer depths in the subtropics will result in more available nutrients to fuel phytoplankton production, but this also leads to lower light levels and cooler sea water temperatures. Likewise, the negative correlation between the nutrient limitation term and NPP in the North Atlantic (Figure 4a) is explained by a strong positive correlation with the light limitation term; nutrients are more limiting during high light conditions associated with increased stratification.

These correlations between the phytoplankton limitation terms and NPP anomalies help to explain why NPP is potentially predictable in some oceanic regions and not in others. Regions where NPP is driven by nutrient availability, in general, have especially high potential predictability (Figures 3 and 4d). This is in agreement with Séférian et al. (2014), who found that nutrient advection plays an important role in predicting NPP in the tropical Pacific. One obvious exception is the northern Indian Ocean, where NPP is controlled by nutrient input, but predictability is low. Interestingly, chlorophyll is predictable in the northern Indian Ocean (Figure S4), highlighting the differences between chlorophyll and NPP potential predictability in the CESM-DPLE. Areas of the ocean in which variability in NPP is driven predominantly by temperature are more limited in spatial extent; these have moderate to low potential predictability 1 year in advance (CESM-DPLE NPP predictions in primarily temperature-driven regions do not show a significant improvement over a persistence forecast; Figures 3 and 4). While we do observe a few light-driven regions with high potential predictability (e.g., western subpolar North Atlantic), most areas where variability in NPP is driven by light (e.g., subpolar Southern Ocean) are not predictable even 1 year in advance. Indeed, interannual variations in PAR at the surface are not predictable in most regions (Figures 5g, 5h, and S5) and mixed layer depth is predictable for one lead year in many oceanic regions but not in the Southern Ocean (Figures 5j-5l and S6). This explains why we observe low potential predictability in the light-controlled Southern Ocean; we hypothesize that variations in cloud cover or sea ice may be contributing to the unpredictability in surface PAR. In contrast, the potential predictability of the physical flux of PO_4 (as a representative nutrient; see section 2) is high 1 year in advance for many areas of the ocean (Figures 5a-5c and S7). SST also shows high potential predictability for lead times of several years (Figures 5d-5f), especially in the North Atlantic, where SST ACC is high for forecast lead times >9 years (Figure S8; Yeager et al., 2018). The parts of the North Atlantic where temperature is the primary driver are therefore quite predictable (e.g., the region north of Iceland). In contrast, the areas of the North Atlantic where there was no dominant driver show lower predictability (white areas on Figure 4d). There are, however, some patches of predictable NPP in the light-driven regions of the North Atlantic subpolar gyre (yellow areas on Figure 4d), indicating that mixed layer depth (which does show high ACC in this region; Figures 5j–5l and S7) may be the primary driver of light variations in this region.

Interestingly, there are some regions where NPP is potentially predictable even though the potential predictability of bottom-up drivers is quite low. This includes parts of the western Atlantic and southeast Pacific for lead times >1 year (Figures 3 and 5). This points to another potential driver of NPP: the persistence of phytoplankton biomass from 1 year to the next within a region. In low productivity regions where nutrients are efficiently recycled in the upper ocean, initialization of biomass anomalies in the CESM-DPLE forecasts may be contributing to predictability by initializing how much biomass (and the nutrients contained therein) is in a region. Places and forecast lead times where NPP is potentially predictable despite low predictability of bottom-up drivers are typically oligotrophic systems, and we speculate that biomass initialization is enabling this predictability. To assess how much biomass persistence could contribute to predicting NPP, we calculated a persistence forecast for phytoplankton biomass using the CESM-FOSI reconstruction; phytoplankton biomass indeed has high persistence in the above-mentioned regions (Figure S9). Though we did not assess the potential predictability of phytoplankton biomass using the CESM-DPLE, chlorophyll predictability in these regions is high for several lead years (Figure S4), suggesting that biomass could also gain additional predictability over persistence using the CESM-DPLE forecasting system. This helps to explain why NPP may maintain predictability even though its physical drivers do not. Another specific example of this phenomenon can be seen in the Gulf of Mexico LME, where marine NPP is primarily controlled by nutrient availability (Figure 4). However, neither physical supply of nutrients nor SST is predictable here, while light has low to moderate predictability (Figure 5), and yet NPP has fairly consistent potential predictability out to forecast lead Year 10 (Figures 3 and S3). Here persistence of phytoplankton biomass and efficient nutrient recycling may be helping maintain long-term predictability. In the following section we further explore the predictability of marine NPP in the world's LMEs.

3.4. NPP Predictability in LMEs

Assessing potential predictability of marine NPP in LMEs could be useful to societies dependent on living marine resources. LMEs are primarily coastal regions, and thus, the physical dynamics that influence the drivers of marine NPP could be difficult to adequately simulate given the coarse (nominal 1°) resolution of CESM. However, we do observe predominately positive correlations between modeled NPP from the CESM-FOSI reconstruction and satellite-derived NPP over the MODIS satellite record, yet variability in these regions is generally underestimated (Figure S10). Therefore, despite the model's coarse resolution and an underestimation of NPP variability, NPP prediction using the CESM-DPLE may help to inform the direction of interannual anomalies of marine NPP in LMEs. Potential predictability for all LMEs is presented in Table S1 for forecast lead Years 1 through 5. In this section, we invoke NPP correspondence with satellite-derived NPP (which admittedly contains its own uncertainties) in combination with potential predictability to assess how useful NPP forecasts may be for various LMEs.

In general, as with open ocean regions, LMEs in oceanic regions where NPP is driven by variations in nutrient availability have higher potential predictability than those driven by light/temperature (Figure 6, top half of LMEs in panel (m)). One of these is the California Current LME (Figures 6a and 6g), an economically important region for fisheries. Here, wind-driven upwelling brings nutrients to the surface, stimulating production by phytoplankton (Rykaczewski & Checkley, 2008); NPP is simulated well by the CESM in the California Current (Figure S10), and potential predictability of NPP is significantly higher than a persistence forecast for a lead Year 1 forecast (Figures 6a, 6g, and S11), indicating that CESM-DPLE NPP predictions may be useful here. Likewise, NPP in two other nutrient-driven LMEs, the Insular Pacific-Hawaiian LME and the Agulhas Current, has good correspondence to satellite-derived NPP (Figure S10), and forecasts show particularly good improvement over a persistence forecast (Figures 6b, 6h, 6m, and S11). Also noteworthy, NPP in the northeast and east central Australian Shelf regions has high potential predictability (Figure 6m) and moderate correspondence to satellite-derived NPP. These two Australian LMEs contain the Great Barrier Reef, and therefore, high NPP predictability could be relevant to conservation efforts. Other nutrient-driven



Global Biogeochemical Cycles



Figure 6. Potential predictability of marine NPP in Large Marine Ecosystems (LMEs). Panels (a)–(f) show time series of NPP anomalies from the CESM-FOSI NPP reconstruction (black), the lead Year 1 forecast (red), and the satellite-derived NPP (blue) for six example LMEs; red and blue *r* values in the upper right of each plot refer to the correlation between the CESM-FOSI reconstruction and a LY1 forecast and the CESM-FOSI reconstruction and the satellite-derived NPP, respectively. Note that the right *y* axis in panel (f) is on a different scale and used to plot the satellite-derived NPP. Panels (g)–(l) show potential predictability (ACC) as a function of forecast lead year for the CESM-DPLE initialized forecast (red) and a persistence forecast (turquoise) for the same example LMEs as in the first column. Panel (m) shows the potential predictability (ACC) for all LMEs for forecast lead Years (LY) 1, LY2, and LY3, ordered with nutrient-driven regions on the top and temperature/light-driven regions on the bottom half. Black dots in the second column of plots and stars in panel (m) indicate a significant improvement in ACC over persistence.

LMEs (e.g, Gulf of California, southeast U.S. Shelf, and the Kuroshio current) have high potential predictability for at least 1 year in advance, but NPP in these regions from the CESM-FOSI reconstruction does not correlate highly with satellite-derived NPP (Figures S10 and S11).

Some temperature and light-driven LMEs have significantly higher potential predictability than persistence for at least one lead year, though these generally have lower ACC than nutrient-driven regions (Figures 6 and S11). One of these is the Gulf of Alaska, where an ACC of \sim 0.5 denotes a large improvement over a

persistence (Figures 6d, 6j, and S11); a persistence forecast for NPP appears to be particularly inadequate in this variable region. Potential predictability of mixed layer depth in the Gulf of Alaska (Figures 5j–5l) may be enabling this improvement over persistence. Other similar LMEs include the southeast Australian Shelf, New Zealand shelf, Yellow Sea, Beaufort Sea, Patagonian Shelf, and the North Sea (Figure 6m, bottom half of LMEs). NPP anomalies from the CESM-FOSI reconstruction show positive correlations with both light and temperature in these regions (Figure 4), but since surface PAR is not particularly predictable in these regions, we attribute most the potential predictability in these LMEs to temperature-driven variations in NPP and mixed layer depth anomalies. However, NPP variations from the CESM-FOSI reconstruction in most of these high-latitude LMEs do not correlate highly with satellite-derived NPP observations, limiting their applicability.

Many LMEs show modestly improved potential predictability over a persistence forecast during the first several forecast lead years (Figure 6m, regions without stars). The Gulf of Thailand LME shows the highest ACC of all the LMEs (ACC = 0.94), but surprisingly, the CESM-DPLE forecast is not significantly better than persistence for a lead Year 1 forecast (see Figures 6c and 6i). In this region the interannual variability in marine NPP is quite low compared to decadal variability (Figure 6c) and therefore initialized forecasts only show only a slight advantage over persistence for a lead Year 1 forecast, although this advantage widens for subsequent forecast lead years (Figure 6i). As mentioned above, the Gulf of Mexico shows persistence in phytoplankton biomass which, along with efficient nutrient recycling, may help drive NPP potential predictability. Though we see only a small improvement using a CESM-DPLE NPP forecast over persistence in this region, we do note that CESM-FOSI-simulated NPP in the Gulf of Mexico (as well as the Caribbean LME) corresponds well to satellite-derived NPP (Figure S10) suggesting that forecasts here may still be somewhat useful. Lastly, though overall NPP variability in the Canary Current LME is underestimated (Figures 6f and S10b), interannual variations correlate quite well with satellite-derived NPP (Figures 6 and S10a) meaning that predictions could help inform the direction of NPP anomalies in this region. Potential predictability of NPP in the Canary Current is higher than a persistence forecast, but the improvement is only significant for forecast lead Year 2 (Figures 6f and 6l).

Finally, while the CESM-DPLE offers improved forecasts (over persistence) in many LMEs, there are regions where potential predictability using the CESM-DPLE is always low. These include the Benguela Current LME, the northwest Indian Ocean LMEs, and Antarctic and Arctic LMEs. In a few regions a persistence forecast is simply superior. In fact, the Benguela Current is the only LME where a persistence forecast has significantly better ACC than the CESM-DPLE forecast (Figure 3). There are also places where NPP is difficult to predict whether using an initialized forecast system, such as the CESM-DPLE, or a persistence forecast. Predicting NPP in high-latitude LMEs (e.g., Arctic LMEs), where temperature and light are controlling factors on NPP, is difficult; ACC is usually somewhat higher using the CESM-DPLE than persistence but still too low to be useful. These regions highlight the limitations of ESM-based prediction systems for forecasting interannual anomalies of NPP in the oceans.

4. Conclusions

Here we have demonstrated that initialized Earth system prediction is capable of forecasting interannual variations in marine NPP for many regions of the ocean. We show that the principle drivers of marine NPP variability influence its potential predictability and that nutrient-driven regions are, in general, more predictable than light- and temperature-driven regions. As marine NPP quantifies converted energy that is available to the rest of marine ecosystems, including living marine resources, we demonstrate NPP potential predictability in society-relevant regions, the world's LMEs. We find that initialized prediction using the CESM-DPLE offers an improvement over persistence in some LMEs and many open ocean regions for predicting NPP anomalies at least 1 year in advance.

Making decadal forecasts of NPP using a ESM prediction system, such as the CESM-DPLE, carries several caveats. The coarse resolution of the model lacks the simulation of mesoscale currents which can be important for variations of NPP in some regions (Harrison et al., 2018), as well as coastal upwelling processes (Brady et al., 2019; Siedlecki et al., 2016; Small et al., 2015). Additionally, marine ecosystems in CESM are represented in a simplified way, simulating only three PFTs and one zooplankton, while lacking upper trophic levels (Moore et al., 2013). This simplification could make translating NPP predictions to predictions of living marine resource availability more challenging. One future direction of this research would be to link NPP with fishery yields as Park et al. (2019) have done with chlorophyll and SST prediction. While we showed NPP predictability in LMEs, the link between NPP and productivity of living marine resources is not always straightforward (see, e.g., Friedland et al., 2012), complicated by uncertainties around fish catch estimates, variable fishing technology and effort, and diverse marine food web dynamics. For instance, Stock et al. (2017) developed an energy-based model that combined NPP with other ecosystem and management characteristics (e.g., benthic versus pelagic energy pathways and fishing effort) to explain fish catch in LMEs better than a model that uses NPP alone. In any case, NPP, as the base of the marine food web, remains an important metric for fishery management and the basis for most fish biomass models (Lotze et al., 2019). Therefore, improving the simulation of marine NPP in ESMs would add utility to NPP forecasts. Such efforts could entail the incorporation of river nutrient inputs into coastal regions or refinement of atmospheric forcing.

This study contributes to a growing body of research using marine ecological forecast products. While ocean physics forecasts are more widespread (e.g., Stock et al., 2015; Yeager et al., 2012), marine ecological forecasting is still in the early stages (Park et al., 2019; Payne et al., 2017; Tommasi et al., 2017). NPP is a complex variable to predict, requiring the adequate simulation and parameterization of physical drivers, as well as ecological processes. We build on regional predictability studies (e.g., Séférian et al., 2014) by evaluating NPP prediction across the open oceans, as well as in coastal LMEs. This study provides a benchmark of using an initialized Earth system prediction system in forecasting interannual fluctuations of NPP in the world's oceans.

References

- Behrenfeld, M. J., Boss, E., Siegel, D. A., & Shea, D. M. (2005). Carbon-based ocean productivity and phytoplankton physiology from space. *Global Biogeochemical Cycles*, 19, 1–14. https://doi.org/10.1029/2004GB002299
- Behrenfeld, M. J., & Falkowski, P. G. (1997). Photosynthetic rates derived from satellite-based chlorophyll concentration. *Limnology and Oceanography*, 42(1), 1–20. https://doi.org/10.4319/lo.1997.42.1.0001
- Behrenfeld, M. J., O'Malley, R. T., Boss, E. S., Westberry, T. K., Graff, J. R., Halsey, K. H., & Brown, M. B. (2015). Revaluating ocean warming impacts on global phytoplankton. *Nature Climate Change*, 6, 323–330. https://doi.org/10.1038/nclimate2838
- Brady, R. X., Lovenduski, N. S., Alexander, M. A., Jacox, M., & Gruber, N. (2019). On the role of climate modes in modulating the air-sea CO₂ fluxes in eastern boundary upwelling systems. *Biogeosciences*, *16*(2), 329–346. https://doi.org/10.5194/bg-16-329-2019
- Bruno, J. F., Bates, A. E., Cacciapaglia, C., Pike, E. P., Amstrup, S. C., Van Hooidonk, R., & Aronson, R. B. (2018). Climate change threatens the world's marine protected areas. *Nature Climate Change*, 8(6), 499–503. https://doi.org/10.1038/s41558-018-0149-2
- Edgar, G. J., Stuart-Smith, R. D., Willis, T. J., Kininmonth, S., Baker, S. C., Banks, S., & Thomson, R. J. (2014). Global conservation outcomes depend on marine protected areas with five key features. *Nature*, 506, 216–220. https://doi.org/10.1038/nature13022

Eppley, R. W. (1972). Temperature and phytoplankton growth in the sea. Fishery Bulletin, 70(4), 1063–1085.

- Falkowski, P. (2012). Ocean science: The power of plankton. Nature, 483(7387), S17–S20. https://doi.org/10.1038/483S17a
- Friedland, K. D., Stock, C., Drinkwater, K. F., Link, J. S., Leaf, R. T., Shank, B. V., & Fogarty, M. J. (2012). Pathways between primary production and fisheries yields of largemarine ecosystems. *PLOS ONE*, 7(1), 1–11. https://doi.org/10.1371/journal.pone.0028945
- Harrison, C. S., Long, M. C., Lovenduski, N. S., & Moore, J. K. (2018). Mesoscale effects on carbon export: A global perspective. Global Biogeochemical Cycles, 32, 680–703. https://doi.org/10.1002/2017GB005751
- Kay, J. E., Deser, C., Phillips, A., Mai, A., Hannay, C., Strand, G., & Vertenstein, M. (2015). The Community Earth System Model (CESM) large ensemble project: A community resource for studying climate change in the presence of internal climate variability. *Bulletin of the American Meteorological Society*, 96(8), 1333–1349. https://doi.org/10.1175/BAMS-D-13-00255.1
- Krumhardt, K. M., Lovenduski, N. S., Long, M. C., & Lindsay, K. (2017). Avoidable impacts of ocean warming on marine primary production: Insights from the CESM ensembles. *Global Biogeochemical Cycles*, 31, 114–133. https://doi.org/10.1002/2016GB005528
- Laufkötter, C., Vogt, M., Gruber, N., Aita-Noguchi, M., Aumont, O., Bopp, L., & Völker, C. (2015). Drivers and uncertainties of future global marine primary production in marine ecosystem models. *Biogeosciences*, 12(23), 6955–6984. https://doi.org/10.5194/bg-12-6955-2015
- Li, H., Ilyina, T., Müller, W. A., & Landschützer, P. (2019). Predicting the variable ocean carbon sink. *Science Advances*, 5, 4. https://doi. org/10.1126/sciadv.aav6471
- Long, M. C., Lindsay, K., Peacock, S., Moore, J. K., & Doney, S. C. (2013). Twentieth-century oceanic carbon uptake and storage in CESM1(BGC). *Journal of Climate*, 26(18), 6775–6800. https://doi.org/10.1175/JCLI-D-12-00184.1
- Lotze, H. K., Tittensor, D. P., Bryndum-Buchholz, A., Eddy, T. D., Cheung, W. W. L., Galbraith, E. D., & Worm, B. (2019). Global ensemble projections reveal trophic amplification of ocean biomass declines with climate change. *Proceedings of the National Academy of Sciences*, 116(26), 12,907–12,912. https://doi.org/10.1073/pnas.1900194116
- Lovenduski, N. S., Yeager, S. G., Lindsay, K., & Long, M. C. (2019). Predicting near-term variability in ocean carbon uptake. Earth System Dynamics, 10(1), 45–57. https://doi.org/10.5194/esd-10-45-2019
- Meehl, G. A., Goddard, L., Boer, G., Burgman, R., Branstator, G., Cassou, C., & Yeager, S. (2014). Decadal climate prediction: An update from the trenches. *Bulletin of the American Meteorological Society*, *95*(2), 243–267. https://doi.org/10.1175/BAMS-D-12-00241.1
- Moore, J. K., Doney, S. C., & Lindsay, K. (2004). Upper ocean ecosystem dynamics and iron cycling in a global three-dimensional model. *Global Biogeochemical Cycles*, 18, GB4028. https://doi.org/10.1029/2004GB002220
- Moore, J. K., Lindsay, K., Doney, S. C., Long, M. C., & Misumi, K. (2013). Marine ecosystem dynamics and biogeochemical cycling in the Community Earth System Model [CESM1(BGC)]: Comparison of the 1990s with the 2090s under the RCP4.5 and RCP8.5 scenarios. Journal of Climate, 26(23), 9291–9312. https://doi.org/10.1175/JCLI-D-12-00566.1

Morel, A. (1991). Light and marine photosynthesis: A spectral model with geochemical and climatological implications. *Progress in Oceanography*, *26*(3), 263–306. https://doi.org/10.1016/0079-6611(91)90004-6

Acknowledgments

We are grateful for funding from the National Science Foundation (OCE-1752724 and OCE-1558225). The CESM-DPLE was generated using computational resources provided by the National Energy Research Scientific Computing Center, which is supported by the Office of Science of the U.S. Department of Energy under Contract DEAC02-05CH11231, as well as by an Accelerated Scientific Discovery Grant for Chevenne (https:// doi.org/10.5065/D6RX99HX) that was awarded by NCAR's Computational and Information Systems Laboratory. The NCAR contribution to this study was supported by the National Oceanic and Atmospheric Administration Climate Program Office under Climate Variability and Predictability Program Grant NA09OAR4310163 and the National Science Foundation Collaborative Research EaSM2 Grant OCE-1243015. This material is based upon work supported by the National Center for Atmospheric Research, which is a major facility sponsored by the National Science Foundation under Cooperative Agreement 1852977. The CESM simulation data analyzed in this paper are available from the project web page of the CESM Decadal Prediction Large Ensemble (http://www.cesm.ucar.edu/projects/ community-projects/DPLE/). We used the Climpred Python analysis package to do some of the analyses shown here (https://github.com/bradyrx/climpred. git). The authors thank Elizabeth Maroon for help with processing CESM-DPLE data sets and Andrew Ross from NOAA for a helpful review of this manuscript.

- Park, J. Y., Stock, C. A., Dunne, J. P., Yang, X., & Rosati, A. (2019). Seasonal to multiannual marine ecosystem prediction with a global Earth system model. *Science*, *365*(6450), 284–288. https://doi.org/10.1126/science.aav6634
- Pauly, D., Alder, J., Booth, S., Cheung, V., Christensen, V., Close, C., & Zeller, D. (2008). The UNEP large marine ecosystems report: A perspective on changing conditions in LMEs of the world's regional seas. In K. Sherman, & G. Hempel (Eds.). Nairobi, Kenya: UNEP.
- Pauly, D., & Christensen, V. (1995). Primary production required to sustain global fisheries. Nature, 374(6519), 255–257. https://doi.org/ 10.1038/374255a0
- Payne, M. R., Hobday, A. J., MacKenzie, B. R., Tommasi, D., Dempsey, D. P., Fässler, S. M. M., & Villarino, E. (2017). Lessons from the first generation of marine ecological forecast products. Frontiers in Marine Science, 4, 289. https://doi.org/10.3389/fmars.2017.00289
- Rykaczewski, R. R., & Checkley, D. M. (2008). Influence of ocean winds on the pelagic ecosystem in upwelling regions. Proceedings of the National Academy of Sciences, 105(6), 1965–1970. https://doi.org/10.1073/pnas.0711777105
- Saba, V. S., Friedrichs, M. A. M., Antoine, D., Armstrong, R. A., Asanuma, I., Behrenfeld, M. J., & Westberry, T. K. (2011). An evaluation of ocean color model estimates of marine primary productivity in coastal and pelagic regions across the globe. *Biogeosciences*, 8(2), 489–503. https://doi.org/10.5194/bg-8-489-2011
- Séférian, R., Bopp, L., Gehlen, M., Swingedouw, D., Mignot, J., Guilyardi, E., & Servonnat, J. (2014). Multiyear predictability of tropical marine productivity. Proceedings of the National Academy of Sciences, 111(32), 11646. https://doi.org/10.1073/pnas.1315855111
- Sherman, K. (2005). 1—The large marine ecosystem approach for assessment and management of ocean coastal waters. In T. M. Hennessey, & J. G. Sutinen (Eds.), Sustaining large marine ecosystems (Vol. 13, pp. 3–16). Amsterdam: Elsevier. https://doi.org/10.1016/ S1570-0461(05)80025-4
- Sherman, K. (2014). Toward ecosystem-based management (EBM) of the world's large marine ecosystems during climate change. *Environmental Development*, 11, 43–66. https://doi.org/10.1016/j.envdev.2014.04.006
- Sherman, K., Belkin, I. M., Friedland, K. D., O'Reilly, J., & Hyde, K. (2009). Accelerated warming and emergent trends in fisheries biomass yields of the world's large marine ecosystems. AMBIO: A Journal of the Human Environment, 215-224, 38(4). https://doi.org/10.1579/ 0044-7447-38.4.215
- Siedlecki, S. A., Kaplan, I. C., Hermann, A. J., Nguyen, T. T., Bond, N. A., Newton, J. A., & Feely, R. A. (2016). Experiments with seasonal forecasts of ocean conditions for the northern region of the California Current upwelling system. *Scientific Reports*, 6, 27203 EP-. https://doi.org/10.1038/srep27203
- Small, R. J., Curchitser, E., Hedstrom, K., Kauffman, B., & Large, W. G. (2015). The Benguela upwelling system: Quantifying the sensitivity to resolution and coastal wind representation in a global climate model. *Journal of Climate*, 28(23), 9409–9432. https://doi.org/10.1175/ JCLI-D-15-0192.1
- Smith, D. M., Cusack, S., Colman, A. W., Folland, C. K., Harris, G. R., & Murphy, J. M. (2007). Improved surface temperature prediction for the coming decade from a global climate model. *Science*, 317(5839), 796–799. https://doi.org/10.1126/science.1139540
- Stock, C. A., John, J. G., Rykaczewski, R. R., Asch, R. G., Cheung, W. W. L., Dunne, J. P., & Watson, R. A. (2017). Reconciling fisheries catch and ocean productivity. *Proceedings of the National Academy of Sciences*, 114(8), E1441–E1449. https://doi.org/10.1073/pnas.1610238114
- Stock, C. A., Pegion, K., Vecchi, G. A., Alexander, M. A., Tommasi, D., Bond, N. A., & Yang, X. (2015). Seasonal sea surface temperature anomaly prediction for coastal ecosystems. *Progress in Oceanography*, *137*, 219–236. https://doi.org/10.1016/j.pocean.2015.06.007
- Tommasi, D., Stock, C. A., Hobday, A. J., Methot, R., Kaplan, I. C., Eveson, J. P., & Werner, F. E. (2017). Managing living marine resources in a dynamic environment: The role of seasonal to decadal climate forecasts. *Progress in Oceanography*, *152*, 15–49. https://doi.org/10. 1016/j.pocean.2016.12.011
- Westberry, T., Behrenfeld, M. J., Siegel, D. A., & Boss, E. (2008). Carbon-based primary productivity modeling with vertically resolved photoacclimation. *Global Biogeochemical Cycles*, 22, 215-224. https://doi.org/10.1029/2007GB003078
- Yeager, S. G., Danabasoglu, G., Rosenbloom, N. A., Strand, W., Bates, S. C., Meehl, G. A., & Lovenduski, N. S. (2018). Predicting near-term changes in the Earth system: A large ensemble of initialized decadal prediction simulations using the community earth system model. *Bulletin of the American Meteorological Society*, 99(9), 1867–1886. https://doi.org/10.1175/BAMS-D-17-0098.1
- Yeager, S. G., Karspeck, A. R., & Danabasoglu, G. (2015). Predicted slowdown in the rate of Atlantic sea ice loss. Geophysical Research Letters, 42, 10,704–10,713. https://doi.org/10.1002/2015GL065364
- Yeager, S., Karspeck, A., Danabasoglu, G., Tribbia, J., & Teng, H. (2012). A decadal prediction case study: Late twentieth-century North Atlantic Ocean heat content. *Journal of Climate*, 25(15), 5173–5189. https://doi.org/10.1175/JCLI-D-11-00595.1